Author's response to reviews

Title: Spontaneous Corneal Melting in Pregnancy: A case report

Authors:

Sudesh K Arya (aryasudesh@yahoo.co.in)
Archana Malik (drarchanaq2002@yahoo.com)
Sonika Gupta (sonika_gupta@yahoo.com)
Hemlata Gupta (hemlatagupta@rediffmail.com)
Sunandan Sood (soods32@yahoo.co.in)

Version: 2 Date: 29 August 2007

Author's response to reviews:

GOVERNMENT MEDICAL COLLEGE & HOSPITAL, CHANDIGARH
DEPARTMENT OF OPHTHALMOLOGY

To,

Editor-in chief
Journal of Medical Case Reports

Subject: Submission of Manuscript titled "Spontaneous Corneal Melting in Pregnancy: A case report."

Dear Sir,

Please find enclosed a manuscript titled "Spontaneous Corneal Melting in Pregnancy: A case report." for your kind consideration for publication in your prestigious journal under category "Case reports." The following change has been made in the original manuscript and it has been marked in red font (Rheumatoid factor, antinuclear antibody, anti-cytoplasmic and anti-DNA antibodies were negative. Patient had not used any topical drops during the intervening period). No grants or sponsorships have been requisitioned for this study. The authors do not have any proprietary or financial interest in any procedure or product mentioned in this manuscript. We are interested in publishing this article in your journal as it is a highly qualified and reputed journal with online submission and free online availability with international readership. I hope this article will find favor with reviewers for publication. We will be able to publish this article with you only if we get a complete waiver on article processing charges as we will not be able to afford it. I will be highly grateful if this could be possible. Kindly acknowledge the receipt of these documents.

Thanking you,

Yours sincerely

Archana Malik
Senior Resident
Department of Ophthalmology
Government Medical College & Hospital
Chandigarh, India